HW Due This Week

ffl At the beginning of each lab section, there will be a folder or box at the front of the lab room where you can turn in homework.

ffl After 10 minutes, the box will disappear. If you have not turned your homework in by then, please turn it in after lab to Dr. Junker’s office (232C Baker).

– This allows the TA’s and other students to get on with the lab exercise, without distractions from latecomers to lab.

– Late hw turned in to Dr. Junker will be graded according to the policy in the course syllabus.

– Please do not turn in late HW to a TA. It just has to go to Dr. Junker anyway.

0-0